Author’s response to reviews

Title: Low lymphocyte count and high monocyte count predicts poor prognosis of gastric cancer

Authors:
Fan Feng (surgeonfengfan@163.com)
Gaozan Zheng (136697876@qq.com)
Guoqing Cai (firstchair@fmmu.edu.cn)
Qiao Wang (594100214@qq.com)
Shushang Liu (1437167914@qq.com)
Zhen Liu (doctorliuzhen@msn.com)
Guanghui Xu (xuguanghui8@126.com)
Fei Wang (wangfeidr@foxmail.com)
Man Guo (1246504365@qq.com)
Xiao Lian (lxlx119@126.com)
Hongwei Zhang (zhanghwfmmu@126.com)

Version: 4 Date: 24 Sep 2018

Author’s response to reviews:

Dear Managing Editor:

A revised manuscript has been submitted. I have include main contents of answers to reviewers’ issues into the discussion of the manuscript. I am pleased that it is acceptable for publication in BMC Gastroenterology,

Best wishes,

Hongwei Zhang, M.D., Ph.D.,

Professor of Medicine.
Department of Digestive Surgery, Xijing Hospital,

The Fourth Military Medical University.

127 West Changle Road, 710032,

Xi’an, Shaanxi Province, China.

Tel: +86-029-84771531.

Fax: +86-029-84771531.

Email: zanghwfmmu@126.com